Author's response to reviews

Title: Effects of epidural lidocaine analgesia on labor and delivery: a randomized, prospective, controlled trial

Authors:

Shahram - Nafisi - (nafisish@yahoo.com)
Rinoo V Shah - (rinoo_shah@yahoo.com)

Version: 4 Date: 29 November 2006

Author's response to reviews: see over
Dear Annabel Phillips,

Thanks from your attention to my work and from the wise comments of professor Halpern and professor Eberhart.

1- I consulted with my statistician and he checked the statistical analysis again. There was a writing mistake. I corrected it: … mean of the groups at least 28 minutes, gives power of 80%, and a 2-tailed …
Thanks Dr. Eberhart for his careful point.

2- I described analgesia measurement for both groups.

3-The rate of cervical dilatation is reported in table 4.

4- The incidence of fever (table 4), hypotension, nausea, vomiting, and dural puncture were included (text).

5-The phrase, in my idea changed to Perhaps, and the word, fine deleted. The reference No.15 changed.
Thanks Dr. Halpern for his careful comments.

Kindest regards,
Shahram Nafisi MD